Author's response to reviews

Title: No evidence of enhanced oxidant production in blood obtained from patients with obstructive sleep apnea

Authors:

Izabela Grabska-Kobylecka (izagra@toya.net.pl)
Andrzej Kobylecki (izagra@toya.net.pl)
Piotr Białasiewicz (pbialasiewicz@neostrada.pl)
Maciej Krol (maciej.krol@lft.com.pl)
Golsa Ehteshamirad (golsa_er@yahoo.com)
Marek Kasielski (kasielsm@tlen.pl)
Dariusz Nowak (dnowak@zdn.am.lodz.pl)

Version: 5 Date: 3 November 2008

Author's response to reviews:

To whom it may concern,

In our last revision we tried to change H2O2 and 10-4 to be written in the proper way (without success) with "2" in lower index and -4 in upper index. We changed twice oxidative stresses into oxidative stress. We placed four new commas and few spaces. We also changed a little the end of the discussion. In addition, we put Escherichia coli in italics. We found also two orthographical mistakes ("assessed" and "antioxidants").

Izabela Grabska-Kobylecka with other Authors